Author’s response to reviews

Title: Exercise and reproductive function in polycystic ovary syndrome: protocol of a systematic review

Authors:

Isis dos Santos (isisk52@gmail.com)
Romilson Nunes (romilson.br@hotmail.com)
Gustavo Soares (gmafalido@hotmail.com)
Tecia Maria Maranhão (teciamaranhao@gmail.com)
Paulo Moreira Dantas (isisk2@hotmail.com)

Version: 1 Date: 23 Nov 2017

Author’s response to reviews:

Responding to Reviewer # 1's comment: I made it as duly noted by the same, and I declare my gratitude in reviewing the protocol, as it really helps in understanding the systematic review protocol. The change is marked in the yellow text.

Responding to Reviewer # 2's comment: Thank you for the review made in the protocol and consider the suggestion mentioned for the text to be valid.